Contents lists available at ScienceDirect



# Journal of Cleaner Production



journal homepage: www.elsevier.com/locate/jclepro

# Hydroeconomic modeling for assessing water scarcity and agricultural pollution abatement policies in the Ebro River Basin, Spain



Safa Baccour<sup>a,e</sup>, Jose Albiac<sup>b,c,e,\*</sup>, Taher Kahil<sup>c</sup>, Encarna Esteban<sup>d,e</sup>, Daniel Crespo<sup>a,e</sup>, Ariel Dinar<sup>f</sup>

<sup>a</sup> Department of Agricultural and Natural Resource Economics, CITA-Government of Aragon, 50059 Saragossa, Spain

<sup>b</sup> Department of Economic Analysis, Facultad de Economía y Empresa, Universidad de Zaragoza, 50005 Saragossa, Spain

<sup>c</sup> International Institute for Applied Systems Analysis (IIASA), A-2361 Laxenburg, Austria

<sup>d</sup> School of Social Sciences and Humanities, Universidad de Zaragoza, Teruel, Spain

<sup>e</sup> Instituto Agroalimentario de Aragón (IA2-Universidad de Zaragoza), 50059 Saragossa, Spain

<sup>f</sup> School of Public Policy, University of California, Riverside, USA

# ARTICLE INFO

Handling Editor: Bin Chen

Keywords: Hydroeconomic modeling Nonpoint pollution Droughts Water quality Climate change Abatement policies

# ABSTRACT

Water scarcity and water quality degradation are major problems in many basins across the world, especially in arid and semiarid regions. The severe pressures on basins are the consequence of the intensification of food production systems and the unrelenting growth of population and income. Agriculture is a major factor in the depletion and degradation of water resources, and contributes to the emissions of greenhouse gases (GHG). Our study analyzes water allocation and agricultural pollution into watercourses and the atmosphere, with the purpose of identifying cost-effective policies for sustainable water management in the Ebro River Basin (Spain). The study develops an hydroeconomic model that integrates hydrological, economic and water quality aspects, capturing the main spatial and sectoral interactions in the basin. The model is used to analyze water scarcity and agricultural pollution under normal and droughts conditions, providing information for evaluating mitigation and adaptation policies. Results indicate that drought events increase nitrate concentration by up to 63% and decrease water availability by 42% at the mouth of Ebro River, highlighting the tradeoffs between water quantity and quality. All mitigation and adaptation policies reduce the effects of climate change, improving water quality and reducing GHGs' emissions, thus lowering environmental damages and enhancing social well-being. Manure fertilization and optimizing the use of synthetic fertilizers are important cost-effective policies increasing social benefits in a range between 50 and 160 million Euros. Results show that irrigation modernization increases the efficient use of nitrogen and water, augmenting social benefits by up to 90 million Euros, and enlarging stream flows at the river mouth. In contrast, manure treatment plants reduce private and social benefits even though they achieve the lowest nitrate concentrations. Our study provides insights on the synergies and tradeoffs between environmental and economic objectives. Another finding is that drought conditions decrease the effectiveness of policies, and increase the tradeoffs between water availability and nitrate pollution. The results contribute to the discussion of designing cost-effective policies for the abatement of agricultural polluting emissions into water and the atmosphere.

# 1. Introduction

Water resources are vitally important for both human livelihoods and natural ecosystems. Water withdrawals have risen sharply in the last century, placing massive pressures on water resources and causing severe water scarcity and degradation problems in most river basins worldwide, especially in arid and semiarid regions (Greve et al., 2018; Dasgupta, 2021). These negative impacts are linked to the strong growth in population and income. Climate change is altering precipitation patterns and making extreme weather events more frequent and intense.

https://doi.org/10.1016/j.jclepro.2021.129459

Received 2 June 2021; Received in revised form 19 October 2021; Accepted 21 October 2021 Available online 27 October 2021

0959-6526/© 2021 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

<sup>\*</sup> Corresponding author. Facultad de Economía y Empresa, Universidad de Zaragoza, Gran Via 2, 50005 Zaragoza, Spain, Tel.: +34 639958988; fax: +34 976761996.

*E-mail addresses:* baccour.safa@gmail.com (S. Baccour), maella@unizar.es (J. Albiac), kahil@iiasa.ac.at (T. Kahil), encarnae@unizar.es (E. Esteban), dcrespoe@ cita-aragon.es (D. Crespo), adinar@ucr.edu (A. Dinar).

Drought is one of the most devastating natural disasters, with serious effects like the shortage of freshwater to meet societal requirements (Ahmadi et al., 2019). Water scarcity and water quality degradation are serious global problems. The challenge is to ensure good quality water to fulfill human, environmental, social, and economic demands in order to support sustainable development (UNESCO, 2021; Berthet et al., 2021). Addressing water scarcity and quality is one important topic of the eighth phase of the Intergovernment Hydrological Programme (IHP--VIII), which focuses on "Water Security: Responses to Local, Regional and Global Challenges (2014-2021)". There are critical connections between water availability and water quality (Jury and Vaux, 2005), and both have been associated with human health (Myers and Patz, 2009), food security (Rockström et al., 2009; Simelton et al., 2012) and sustaining natural ecosystems (Poff et al., 1997). This means that water availability and quality should be assessed in a consistent manner to account for the relationships between water availability and quality.

Nonpoint pollution is responsible for 38% of pressures affecting water bodies in Europe, mainly due to agricultural sources such as nitrates and pesticides (European Environment Agency, 2018). Agriculture is a major source of water quality deterioration and GHG emissions to the atmosphere. Both water pollution by nutrients and GHG loads are complex problems arising from excessive use of fertilizers and intensive livestock farming (Bluemling and Wang, 2018). Nitrous oxide (N<sub>2</sub>O) and methane (CH<sub>4</sub>) are a potent GHGs that contribute to the planet global warming (IPCC et al., 2007; Kanter et al., 2017). Rivers receive large quantities of nutrients, which cause water eutrophication and create large hypoxic dead zones in some regions (Breitburg et al., 2018). Parris (2011) highlights that agricultural water quality is a major environmental issue in OECD countries, and it is a relevant matter for policy consideration in all OECD countries.

Protecting water resources and natural ecosystems requires robust institutions, coupled with compelling and enforceable water policies. Sustainable river basin management is a quite challenging task, considering the current scale of global water degradation in basins. The methodologies needed to address this challenge call for a better understanding of water management problems in order to deploy effective and politically viable measures dealing with water scarcity, droughts, climate change and pollution. Sustainable management of water resources for different uses will not only depend on water quantity withdrawals, but also on nutrient loads, organic matter, salinity, water temperature, and other pollutants (Van Vliet et al., 2017; Barbieri et al., 2019).

The use of hydroeconomic modeling is increasing, driven by the advances of integrating hydrology, environment, and socio-economics in the analysis of water resources management. Several studies investigate the problem of water allocation among sectors using hydro-economic modeling to assess water policies (Ringler et al., 2006; Kahil et al., 2015, 2016a, 2016a; Escriva et al., 2018). Other studies emphasize sectoral and spatial interactions in catchment areas (Bekchanov et al., 2015; Kahil et al., 2016b, 2018; Dogan et al., 2018; Crespo et al., 2019). Despite the widespread use of hydroeconomic modeling in assessing water allocation, the inclusion of the policy analysis for the abatement of water pollution is limited.

The inclusion of water quality is a topic of growing relevance, although there are few studies analyzing water quality by using hydroeconomic modeling. Some examples are salinity pollution (Cai et al., 2003; Aein and Alizadeh, 2021), arsenic in drinking water (Ward and Pulido, 2008), organic matter loads (Moraes et al., 2010), biochemical oxygen demand (Gunawardena et al., 2018), nitrate pollution (Carolus et al., 2020), and environmental and salinity damages in terms of water savings, replacement costs or crop production damages (Booker and Young, 1991, 1994; Brown et al., 1990; Cai et al., 2002; Divakar et al., 2011). Recreation benefits such as boating and fishing are sometimes included in relation to stream flows evaluation, and travel cost or contingent valuation techniques are used for valuation of the ecosystem services (Ringler et al., 2004; Babel et al., 2005; Booker et al., 2005; Ringler and Cai, 2006; Ward and Pulido, 2008, 2012, 2012). In Spain, there are only a couple of previous studies on water pollution abatement using hydroeconomic modeling, where the modeling framework was applied to a hypothetical groundwater system (Peña-Haro, 2009, 2011, 2011).

Some studies assess the tradeoff between water quantity and quality using a simulation model (Yang et al., 2015). However, the tradeoffs between water scarcity and water quality degradation using an optimization model remain unsettled in the literature. The advantage of using an optimization model is in the capacity of the model to maximize the economic benefits under water scarcity and agricultural nonpoint pollution simultaneously, which involves a more realistic approach. This integrated hydroeconomic model is designed to find the most cost-effective management policies (Heinz et al., 2007) and to make socially optimal policy decisions (Gunawardena et al., 2018). The assessment of the relationship between water quantity and quality is important to strengthen hydroeconomic modeling, in order to understand and realize its full power to inform critical policy debates.

In this paper, an integrated hydroeconomic model is developed addressing both water allocation and agricultural nonpoint pollution, with the purpose of looking at the tradeoffs between water quantity and water quality under normal and drought conditions. The model estimates agricultural pollution impacts on both the watercourses (nitrates) and the atmosphere (nitrous oxide and methane). The integration of hydrological, economic and environmental components captures the interactions among components. This provides a better assessment of water allocation options among sectors and spatial locations, showing the large negative impacts of droughts on the system.<sup>1</sup>

Selected climate change mitigation and adaptation policies are evaluated under normal climate and severe drought conditions in order to identify the effectiveness and robustness of policies. These policies could boost the efficient use of nitrogen and water in agricultural activities, reduce pollution loads and improve water and air quality, or protect environmental flows. The hydroeconomic model is developed to analyze the Ebro River Basin in northeastern nearly all basins in Spain are under mounting scarcity pressures and water quality problems that require effective policy intervention (Lassaletta et al., 2009). Climate change and agricultural nonpoint pollution problems have to be tackled locally, with practical alternatives addressing water depletion and pollution.

This study contributes to the literature performing a detailed concurrent assessment of water allocation and pollution abatement solutions at river basin level, using hydroeconomic modeling. The study analyzes how to achieve a more sustainable management of the Ebro Basin, but also contributes to the scientific debate on sustainable policies and measures for water management worldwide. The results of this paper highlight the strong links between water quality and water quantity in the basin, and show that drought conditions reduce water availability and dilution processes, increasing nitrate concentration in water media. Our results indicate also that mitigation and adaptation policies have a double effect by abating pollution into the atmosphere and in watercourses, thus reducing environmental damages and enhancing social welfare.

The paper is organized as follows. Section 2 presents a general description of the study area and the main economic activities in the basin. Section 3 explains the development of the integrated hydroeconomic model for the Ebro basin. Section 4 describes the main results of the mitigation and adaptation policies, and the drought impact in

<sup>&</sup>lt;sup>1</sup> Costs of drought damages have been estimated at \$8 billion per year in the United States (NOAA, 2021), and around 9 billion  $\notin$  per year in the European Union (Cammalleri et al., 2020). Hernández et al. (2013) estimate the cost of the 2005 drought in the Ebro basin at 0.5% of GDP. The evidence during recent years indicates that the drought anomaly in Europe is unprecedented (Büntgen et al., 2021).

water quantity and quality, and section 5 discusses the main findings. Finally, section 6 summarizes the main conclusions.

# 2. The Ebro Basin

The Ebro Basin, located northeast of the Iberian Peninsula, is one of the main European Mediterranean basins. It covers an area of 85,600 km<sup>2</sup>, a fifth of the Spanish territory, and its streamflow is one of the largest in the country. Natural ecosystems of great value cover 30% of the basin area. Precipitation occurs mainly in the Pyrenees, where it exceeds 1000 mm/year, while it does not exceed 350 mm/year in the central part of the basin, where conditions are semi-arid (CHE, 2015). The most important tributaries (Zadorra, Aragon, Gallego, Cinca and Segre) supply the canals of the main irrigation districts and also the most important urban areas in the basin (Fig. 1).

The renewable resources of the Ebro basin are estimated at 14,600  $\text{Mm}^3$ , and withdrawals amount to 8460  $\text{Mm}^3$ , of which 8110  $\text{Mm}^3$  are surface diversions and 350  $\text{Mm}^3$  are groundwater extractions (CHE, 2015). Water use in agricultural activities is estimated at 7680  $\text{Mm}^3$  and urban extractions amounts to 357  $\text{Mm}^3$  supplying three million inhabitants, including households and industries connected to urban networks. The irrigated crops in the Ebro Basin are field crops, fruit trees and vegetables covering an area of 750,000 ha, distributed under surface, sprinkle and drip irrigation technologies (CHE, 2016). The Ebro River is one of Spain's rivers with substantial minimum environmental flows at river mouth. The Ebro water plan of 2015 established the current level of this environmental flow at 3000  $\text{Mm}^3/\text{year}$ .

The Ebro Basin Authority is responsible for water management, water allocation, water quality, and water planning and control. The special characteristic of this institutional approach is the key role played by stakeholders, which are involved at all decision making in the basin governing bodies and in local watershed boards. The Ebro Basin Authority or Confederación Hidrográfica del Ebro (CHE, 2020) indicates that nonpoint pollution represents one of the main pressures on the Ebro coming from agricultural and livestock activities. Almost half surface waters in the basin are being significantly affected, particularly in its middle and lower reaches (Ollero, 2007; Vericat and Batalla, 2006). The mean annual streamflow has decreased 40% in the last 50 years because

of the expansion of irrigation, decreasing rainfall and revegetation (Buendia et al., 2016). The ecological condition of water bodies is threatened by these hydrological alterations and nonpoint pollution loads, impairing the dilution capacity. Herrero et al. (2018) highlight that changes in land use, rainfall, water temperature, and nitrate concentration could lead to a general decrease in the ecosystem quality of water bodies within the basin. Overall, water quality pressures from agricultural nonpoint pollution are degrading the status of water bodies in the Ebro, and require the active intervention of state and federal public authorities together with all water stakeholders.

# 3. The hydroeconomic model

Water is an essential component of sustainable development, underpinning almost all types of economic activities, human water security, and ecosystems services. Challenges to water management such as water scarcity, pollution loads, and the impacts of climate change are threatening human wellbeing and biodiversity. Hydroeconomic analysis is one type of water-economy modeling, which is based on the hydrologic network of river basins. The hydroeconomic approach has clear advantages in evaluating management and policy strategies for adaptation to climate change, by providing efficient water allocations and pollution abatement across water uses and spatial locations. Hydroeconomic models have achieved greater sophistication by integrating agronomic, hydrologic, and economic components (Cai et al., 2003; Harou et al., 2009; Booker et al., 2012; Ward, 2021). This involves a more realistic approach to water allocation and water quality trade-offs across space and sector, and less reliance on temporally and spatially integrated demand functions used by economywide models (Bekchanov et al., 2017).

The hydroeconomic model is used to analyze water allocation among sectors and spatial locations, nonpoint pollution loads across the basin, and also to evaluate drought scenarios and climate change mitigation and adaptation measures. The policy analysis deals with both water allocation adjustments under droughts and climate change, and pollution abatement of nutrient loads and GHG emissions. The model includes the main water uses in the basin: irrigation, livestock, and urban and industrial. Dryland crops are also included in the assessment of



Fig. 1. Map of the Ebro basin.

pollution emissions. The model integrates three components: (1) the hydrological component, (2) the regional economic component, and (3) the environmental component (Fig. 2).

#### 3.1. The hydrological component

The hydrological component is a reduced form hydrological model of the Ebro basin, calibrated with observed stream flows. The reduced form hydrological model is a node-link network, in which nodes represent physical units impacting the stream system, and links represent the connection between these units. The nodes are classified into supply nodes such as rivers, and demand nodes such as irrigation districts, livestock, households and industries. The links could be rivers or canals, and stream flows between supply and demand nodes are characterized by simplified equations using the hydrological concepts of mass balance and continuity of river flows (Kahil et al., 2015). The representation of the interactions among nodes is based on detailed information on each node's spatial location and physical characteristics. The component incorporates information on inflows, withdrawals, return flows and losses, and water metering at selected measurement stations in the basin. The model can simulate the flows at each node and the distribution of water availability between sectors and spatial locations. The hydrologic component is developed using the databases of CHE (2016), and it is calibrated with the observed historical allocations in selected stations of the basin (see Fig. 1S for further details on the Ebro hydrological system). The mathematical formulation is given by the following equations:

$$Wout_{d} = Win_{d} - Wloss_{d} - Div_{d}^{IR} - Div_{d}^{URB} - Div_{d}^{LIV}$$
(1)

$$Win_{d+1} = Wout_d + r_d^{IR} \cdot (Div_d^{IR}) + r_d^{URB} \cdot (Div_d^{URB}) + r_d^{LIV} \cdot (Div_d^{LIV}) + RO_{d+1}$$
(2)

$$Wout_d \ge E_d^{min}$$
 (3)

The first equation shows the mass balance and determines the water outflow  $Wout_d$  in river reach d, which is equal to the inflow  $Win_d$  minus water losses  $Wloss_d$ , and minus the diversions for irrigation  $Div_d^{IR}$ , urban use  $Div_d^{URB}$  and livestock use  $Div_d^{LIV}$ . The second equation guarantees flow continuity in the basin.  $Win_{d+1}$  is the water inflow into the following river reach d + 1 as the sum of the outflow from the upstream water reach  $Wout_d$ , the return flows from upstream irrigation districts  $[r_d^{IR} \cdot (Div_d^{IR})]$ , urban return flows  $[r_d^{URB} \cdot (Div_d^{URB})]$ , livestock return flows  $[r_d^{LIV} \cdot (Div_d^{LIV})]$ , and the runoff entering the river reach from tributaries  $RO_{d+1}$ . The third equation specifies that the water outflow in river reach d must be greater than or equal to the minimum environmental flow imposed on that river reach.

The hydrologic component is calibrated by introducing slack variables in every river reach to balance supply and demand at every node. These variables represent unmeasured water sources or uses. This calibration procedure reproduces the water flows observed in the reference conditions. Water inflows, outflows and characteristics of flow rates in rivers and channels have been taken from databases and reports by CHE (2016) and CEDEX (2020).

# 3.2. The regional economic component

The regional economic component consists of optimization models for irrigation districts, for livestock and dryland crops, and for urban economic surplus. For irrigation, the component is set at irrigation district scale to maximize the benefits of crops subject to a set of technical and resource constraints. Yield functions are linear and decreasing in cropland area, with constant input and output prices. The optimization problem is as follows<sup>2</sup>:

$$Max B_{k}^{IR} = \sum_{ij} C_{ijk}^{'(IR)} \cdot X_{ijk}^{IR}$$
(4)

subject to

$$\sum_{i} X_{ijk}^{IR} \leq Tland_{kj}; \quad i: crop; \ j: flood, \ sprinkler, drip; k$$
  
: irrigation district (5)

$$W_{iik} \cdot X_{iik}^{IR} \leq Twater_k \tag{6}$$

$$\sum_{ij} L_{ijk} \cdot X_{ijk}^{IR} \leq Tlabor_k$$
<sup>(7)</sup>

$$\sum_{ij} N_{ijk} \cdot X_{ijk}^{IR} \leq Tnitrogen_k$$
(8)

$$X_{ijk}^{IR} \geq 0$$
 (9)

where  $B_k^{IR}$  is the private benefit in each irrigation district k and  $C_{ijk}^{(IR)}$  is net income per hectare of crop i using irrigation technology j. The decision variable of the optimization problem is  $X_{ijk}^{IR}$ , the area of crop i with irrigation system j. Irrigated crops are grouped into field crops, vegetables and fruit trees, using surface, sprinkler and drip irrigation systems. Field crops are irrigated by surface and sprinkler irrigation, while vegetables and fruit trees are irrigated by surface and drip irrigation.

Equation (5) is the land constraint and it represents the land available in each irrigation district k equipped with irrigation system j,  $Tland_{kj}$ . Equation (6) is the water constraint and it represents the water available in each irrigation district k,  $Twater_k$ , where  $W_{ijk}$  is the requirement for water per hectare and per crop i with irrigation system j. The level of available water,  $Twater_k$ , is the variable linking the optimization model of the irrigation districts and the hydrological component. Equation (7) is the labor constraint and it represents the labor available in each irrigation district k,  $Tlabor_k$ .  $L_{ijk}$  is the requirement for labor per hectare of crop i with irrigation system j. Equation (8) is the nitrogen constraint and it represents the nitrogen available in each irrigation system j. Equation (8) is the nitrogen district k,  $Tnitrogen_k$ .  $N_{ijk}$  is the nitrogen available in each irrigation system j. Equation (9) is the non-negativity constraint of the crop surface area. Net income per hectare  $C'_{ijk}^{(IR)}$  is the difference between revenues and costs and it is defined as:

$$C_{iik}^{\prime(IR)} = P_i Y_{ijk} - CP_i \tag{10}$$

where  $P_i$  is the price of crop *i*,  $Y_{ijk}$  is the yield of crop *i* under irrigation system *j* in irrigation district *k*, and  $CP_i$  represents the direct and indirect costs of crop *i*.

The Ricardian rent principle is used in the yield function by assuming that yield decreases as the scale of production increases. The yield function is linear and decreasing in the area of crop i under irrigation system j and it is expressed by:

$$Y_{iik} = \beta 0_{iik} + \beta 1_{iik} X_{iik}^{IR} \tag{11}$$

Positive mathematical programming (PMP) is used to calibrate irrigated crop production following the approach of Dagnino and Ward (2012) in order to solve the aggregation and over-specialization problems. The procedure estimates the linear yield function parameters  $\beta 0_{ijk}$  and  $\beta 1_{ijk}$ .

Livestock and dryland cultivation components are set at watershed

<sup>&</sup>lt;sup>2</sup> We use the standard mathematical programing formulation of farm modeling (e.g. Kahil et al., 2015, 2016; Crespo et al., 2019).



Fig. 2. Modeling framework.

board scale, and maximize benefits subject to technical and resource constraints. A constant yield production function for crops and constant input and output prices are used (see A1 in Appendix A of Supplementary Materials for further details).

The economic benefits of urban water use are determined using a social surplus model, by maximizing the consumer and producer surpluses for the main urban centers in the basin, subject to the water supply and demand balance constraint. The optimization problem is expressed as follows:

$$MaxB_{u}^{URB} = \left(a_{du} \cdot Q_{du} - \frac{1}{2} \cdot b_{du} \cdot Q_{du}^{2} - a_{su} \cdot Q_{su} - \frac{1}{2} \cdot b_{su} \cdot Q_{su}^{2}\right)$$
(12)

subject to

$$Q_{du} - Q_{su} \le 0 \tag{13}$$

$$Q_{du}; Q_{su} \ge 0 \tag{14}$$

where  $B_{u}^{URB}$  is the sum of the consumer and producer surpluses in urban center u. The variables  $Q_{du}$  and  $Q_{su}$  are water supply and demand in urban center u, respectively. The parameters  $a_{du}$  and  $b_{du}$  are the intercept and the slope of the inverse demand function,  $P_{du} = a_{du} - b_{du} \cdot Q_{du}$ . The parameters  $a_{su}$  and  $b_{su}$  are the intercept and the slope of the inverse water supply function,  $P_{su} = a_{su} + b_{su} \cdot Q_{su}$ . Equation (13) indicates that

water supply is greater than or equal to demand. The variable  $Q_{su}$  is the quantity of water supplied and it is the variable linking the urban model with the hydrological component. The water demand parameters have been obtained from the estimates by Arbués et al. (2004) and Arbués et al. (2010).

# 3.3. The environmental component: water and atmosphere pollution

Agricultural nonpoint pollution is analyzed in the environmental component, assessing the environmental damage derived from agricultural activities in the Ebro Basin. The impact of nonpoint pollution is assessed by estimating the nitrate loads into watercourses and GHG emissions from irrigated and dryland crops, and from livestock. GHG emissions from cropland include direct and indirect nitrous oxide (N<sub>2</sub>O), while livestock emissions include methane (CH<sub>4</sub>) from enteric fermentation and nitrous oxide and methane from manure management. The environmental component includes the minimum environmental flows at each section of the basin. The estimation of the social costs of agricultural nonpoint pollution is a complex task that requires a detailed analysis of the biophysical processes generating source emissions and transport and fate processes, the damages from water and atmosphere pollution, and the costs of these damages.

In this study, the methodology applied to estimate GHG emissions

from agriculture is the Tier 1 method of the IPCC (2019a; 2019b). The nitrogen pollution is estimated from leaching and runoff from crops, and from the nitrogen excreted by livestock. The biophysical information for each crop and irrigation system are taken from literature reviews and fertilization practices in Spain published by the Spanish Ministry of Agriculture. Emission factors and the data used in the estimation of GHG emissions are taken from IPCC (2019a; 2019b). We assume also that the NO<sub>3</sub>–N loads reaching watercourses are 40% of all nitrogen loads at the source of pollution, and the NO<sub>3</sub>–N loads reaching the Ebro river mouth represent only 10% of all nitrogen loads at the source of pollution. This is based on the results of Lassaletta et al. (2012), which indicate a high level of retention in the basin (90%).

The environmental damage of agricultural activities is the sum of the cost of GHG emissions and the cost of nitrogen pollution into watercourses, and are given by the expression:

$$ED = GHG E \cdot SC + 0.4 \cdot Nload \cdot NC$$
(15)

where the damage of GHG emissions is determined by the volume of GHG emissions (GHG E) and the social cost of carbon (SC) set at 40  $\epsilon$ /tCO<sub>2</sub>e, which is taken from OECD estimates (Smith and Braathen, 2015) which are close to current US EPA regulation (US\$51/tCO<sub>2</sub>e). The environmental damage from nitrates is calculated multiplying the volume of nitrate loads from crops and livestock (Nload), by the cost to removing nitrate from water (NC) at 1.3  $\epsilon$ /kg NO<sub>3</sub>–N (Martínez and Albiac, 2006). Details on calculations are presented in Appendix A of Supplementary Materials.

#### 3.4. Ebro optimization model and model application

The optimization model of the Ebro Basin integrates the three components described above, and the objective function represents social benefits, the sum of private benefits (B) minus environmental damages (ED) (See A3 in Appendix A of Supplementary Materials for further details). The maximization of social benefits covers all water sectors and spatial locations. The optimization problem is given by:

$$Max(B - ED) \tag{16}$$

subject to all hydrological, technical, economic and environmental constraints of irrigated, dryland, and livestock activities. The mathematical programing GAMS package has been used for the Ebro model. The model has been solved using a nonlinear programming algorithm (CONOPT). Ward (2021) indicates that GAMS might be an effective tool for implementing linear, non-linear, and integer optimization. It can solve large systems of non-linear equations simultaneously. The system is flexible, open, and self-documenting, with obvious connections between model formulation and solution.

The hydroeconomic model is used to analyze the interdependence between water quantity and water quality, under normal water inflows and drought scenarios. Drought scenarios are used to understand future drought severity levels, and the ensuing impacts of water scarcity and pollution on social benefits in the basin. Moderate and severe drought scenarios assume reductions of 30% and 40% in water inflows, respectively, relative to the flows under normal climate conditions. Then, the model is used to assess selected mitigation and adaptation policies under normal climate and severe drought conditions.

This assessment highlights the role that policies could play in the abatement of nonpoint pollution in watercourses and the atmosphere, and also in identifying the tradeoffs between water quality and water scarcity. The analysis shows the effectiveness of policies under extreme droughts and the impacts on water use, pollution loads and their environmental damages, and social benefit outcomes. The selected policies are P1: Optimization of nitrogen fertilization (by reducing fertilization to crop requirements); P2: Substitution of synthetic fertilization by organic fertilization; P3: Irrigation modernization; P4: Manure treatment plants, (Table 1).

#### Table 1

1	Description of policies.							
	Policies	Description						
	P1	Efficient use of nitrogen fertilization at crop requirements without impacts on yields.						
	P2	Substitution of synthetic by organic fertilization up to 60% share (from						

	current 27%).
P3	Replacing surface irrigation by more efficient irrigation technologies.

P4 Use of manure treatment technologies to reduce nitrogen emissions.

## Table 2

Agricultural	use of	resources.	pollution	and	benefits	under	drought	scenarios.
- Mariculture		10000000000	ponucion		DOMOIND		arougne	00000000

Climate conditions	Normal	Moderate	Severe		
	flow	drought	drought		
Lond (1000 ha)					
Land (1000 ha)		262	015		
Field and	557	302	315		
Field crops	399	225	184		
Vegetables	36	30	28		
Fruit trees	122	107	103		
Dryland	1194	1194	1194		
Field crops	900	900	900		
Fruit trees	294	294	294		
Livestock (1000 head)					
Swine	12,913	12,913	12,913		
Ovine	2380	2380	2380		
Beef cattle	724	724	724		
Dairy cattle	74	74	74		
Water use (Mm <sup>3</sup> )					
Agriculture					
Irrigated land	3497	2448	2098		
Livestock	55	55	55		
Urban	322	322	322		
Total	3874	2825	2475		
Irrigation system (1000 ha)					
Flood	292	158	129		
Sprinkler	174	120	104		
Drip	91	84	82		
Streamflow at the river mouth	9272	6366	5406		
(Mm <sup>3</sup> )					
Nitrogen emissions (1000 tNO3-N)	)				
At the source	236	227	225		
Entering water bodies	94	91	90		
Nitrate concentration (mg/l NO <sub>3</sub> <sup>-</sup>	)				
Ebro River mouth	11.3	15.8	18.4		
GHG emissions (MtCO2e)					
N <sub>2</sub> O from crops	0.76	0.58	0.54		
CH₄ from Enteric Ferm.	1.92	1.92	1.92		
N <sub>2</sub> O from Manure Manag.	0.85	0.85	0.85		
CH₄ from Manure Manag.	3.62	3.62	3.62		
Total	7.15	6.97	6.93		
Private benefits (Mf)	,		-,		
Agriculture					
Irrigated land	813	739	705		
Drvland	301	241	211		
Livestock	811	811	811		
Urban	1859	1859	1859		
Total	3784	3650	3586		
Environmental damages (MF)	0/01	0000	0000		
Irrigated land	34	22	10		
Dryland	14	14	14		
Livestock	361	361	361		
Total	400	307	204		
Social benefits (ME)	707	37/	374		
Junianted land	770	717	696		
Intigateu Ianu	//9	/1/	107		
Dryland	287	227	197		
LIVESLOCK	450	450	450		
Urban m- t- 1	1828	1828	1859		
Total	3375	3253	3192		

## 4. Results

4.1. Water allocation, and nonpoint pollution under normal and drought scenarios

The results of water allocation, environmental damages and social benefits under the baseline and drought scenarios are presented in Table 2. Under normal climate conditions, the.

social benefits are 3375 M€ and the total water use reaches 3874  $Mm^3$ . The irrigated land covers 557,000 ha of field crops, fruit trees and vegetables. Dryland covers 1,194,000 ha and livestock herds amount to 2769 thousand Livestock Units (LSU). Employment in the basin is 37,000 Annual Work Units (AWU) for irrigated crops, 21,500 AWU for dryland crops, and 34,000 AWU for livestock rearing. Results show that nitrogen emissions at the source are 236,000 tNO<sub>3</sub>-N and GHG emissions are 7.15 MtCO<sub>2</sub>e from agricultural activities, which concentrate in Canal de Urgel, Canal de Bardenas, and the lower sections of the Segre and Gallego tributaries, given the large irrigated cropland and swine herds in these areas (Fig. 3a; Fig. 4). Nitrogen loads entering watercourses in the Ebro are around 94,000 tNO<sub>3</sub>-N, and the nitrate concentration at the river mouth is estimated at 11.3 mg/l NO<sub>3</sub><sup>--</sup> under normal climate (Fig. 3b). The environmental damages from water pollution and GHG

Journal of Cleaner Production 327 (2021) 129459

emissions are 409 M $\in$ , which are subtracted from the farming private benefits in order to calculate social benefits.

Under drought conditions, water allocation to irrigation districts is reduced proportionally to their regular allocation, while water allocation to urban areas and livestock is maintained. Urban areas take priority over any other water use, followed by livestock. In normal weather conditions, animals only use 1% of water withdrawals, and during droughts water is not a limiting factor for livestock. Under moderate drought, water diversions for irrigation are reduced by 30% with private benefits dropping to 739 M€. Moderate drought reduces irrigated acreage by 35%, especially for less efficient irrigation system. GHG emissions and nitrogen pollution at the source are reduced, while the nitrate concentration at the Ebro River mouth increases by 40% due to the reduction of river flows. Under severe drought conditions, water withdrawals for irrigation are reduced proportionally by 40%. Irrigated cropland generates 686 M€ in private benefits using 2098 Mm<sup>3</sup> of water. The irrigated acreage falls almost by half and nitrogen pollution at the source decreases. However, the nitrate concentration at the mouth of river increases by 63%.

The results show that droughts reduce crops with low profitability and high water requirements, and the cropland acreage under less efficient irrigation technologies (Fig. 2S in Appendix B of Supplementary





Fig. 3. Nitrogen emissions at the source and in water bodies at municipal level.



**Fig. 4.** Agricultural GHG emissions in the Ebro Basin at municipal level. Figure shows N<sub>2</sub>O emissions from crops (N<sub>2</sub>O CE), the CH<sub>4</sub> emissions from enteric fermentation (CH<sub>4</sub> EF), and the N<sub>2</sub>O (N<sub>2</sub>O MM) and CH<sub>4</sub> (CH<sub>4</sub> MM) emissions from manure management.

Materials). The drought scenarios illustrate what are the more efficient water and land management options for adaptation to water scarcity, which vary between irrigation districts and respond to factors such as crop diversification, the level of modernization of irrigation systems, and the access to water resources (Fig. 3S in Supplementary Materials). In addition, results highlight the tradeoff between nitrate concentrations and water availability. Nitrate concentrations increase under drought conditions, as the dilution processes worsen driven by water scarcity.

### 4.2. Policy analysis under normal and drought conditions<sup>3</sup>

# 4.2.1. Optimization of nitrogen fertilization

The efficient use of nitrogen fertilization in irrigated and dryland crops in the Ebro Basin is an interesting policy that can reduce nonpoint pollution into the atmosphere and watercourses. This policy increases the profit of crops by 45 M€ while reducing environmental damages by 12 M€, achieving higher social benefits. The increase in private benefits results from the drop of nitrogen fertilization (-39,000 tN) which reduces nitrogen leaching (-7000 tN) and crops N<sub>2</sub>O emissions (-196,000 tCO<sub>2</sub>e). Cultivated area and water withdrawals increase, reducing the streamflow at the Ebro mouth. Nitrate loads at the source in the basin are reduced to 229,000 tNO<sub>3</sub>-N, declining nitrate concentrations at the river mouth by 0.3 mg/l NO<sub>3</sub><sup>-</sup>.

Under drought conditions, despite the reduction of streamflow at the mouth to 5341  $Mm^3$ , this policy still improves water and atmosphere quality by reducing nitrate concentration to  $18.2 \text{ mg/l NO}_3^-$  and GHG emissions to 6.79 MtCO<sub>2</sub>e, compared to drought conditions without policies. The results point out also that the policy under drought reduces nitrate loads at the source to 220,000 tNO<sub>3</sub>-N but increases water withdrawals to 2566  $Mm^3$ . Compared with the policy in normal flow, nitrate concentration at the mouth rises 65%, and the reason is drought decreases water availability and impairs the dilution processes. In both cases, normal and drought conditions, this policy is efficient in mitigating agricultural pollution into the atmosphere and watercourses

(although reductions are moderate), and in enhancing private profits. The policy benefits both farmers and the environment, generating synergies between environmental and economic outcomes (Table 3). However, its implementation requires the training and willingness to cooperate of farmers.

#### 4.2.2. Substitution of synthetic fertilization by organic fertilization

Substituting synthetic fertilization by organic fertilization is also an interesting policy for reducing nonpoint pollution to the atmosphere and water streams, and avoid the high abatement costs of manure treatment plants. Increasing the circular use of manure as fertilizer from the current 27% up to 60% would promote a more sustainable agriculture by reusing nutrients in the soil and preventing pollution. This study assumes that the cost of manure application amounts to  $3.7 \ \text{€/m}^3$  for a distance of 10 km, which includes transport and specialized equipment costs (Daudén et al., 2011). Results show that manure fertilization increases irrigated land to 584,000 ha and water withdrawals to 4031  $\text{Mm}^3$ , reducing streamflow at the river mouth by 112  $\text{Mm}^3$ . This policy increases organic fertilization up to 153,000 tN, while synthetic fertilization declines, achieving a reduction of 300,000 tCO<sub>2</sub>e in.

GHG emissions and 28,000 tNO<sub>3</sub>-N in nitrate loads into watercourses, which decreases nitrate concentration at the Ebro mouth by 32% to 7.7 mg/1 NO<sub>3</sub><sup>-</sup>. Environmental damages decrease by 109 M $\in$  and private benefits increase by 30 M $\in$  because of the cost savings of organic fertilization, augmenting social benefits up to 3531 M $\in$ .

Under drought conditions, the policy abates nitrate loads at the source to 189,000 tNO<sub>3</sub>-N and GHG emissions to 6.81 MtCO<sub>2</sub>e, while water withdrawals amount to 2564 Mm<sup>3</sup>. However, nitrate concentration increases at the river mouth by 39% to 15.7 mg/l NO<sub>3</sub><sup>-</sup> because of the drought lower stream flows. Compared with drought conditions without any policy, manure fertilization improves water and air pollution, lowering environmental damages (-82 M€) and increasing social benefits (+119 M€). This policy entails synergies in reducing both atmosphere and water pollution, and synergies between economic and environmental outcomes under normal and drought conditions. It shows also an acceptable tradeoff between water quantity (streamflow at the mouth) and quality (pollution abatement) (Table 3).

<sup>&</sup>lt;sup>3</sup> Detailed results on the baseline and policy scenarios are presented in Table 1S, Figs. 4S and 5S of the Supplementary Materials

# Table 3

Use of resources, pollution and benefits for each policy under normal and drought conditions.

	Normal flow					Severe drought				
Policies	Without policies	P1	P2	Р3	P4	Without policies	P1	P2	Р3	P4
Land (1000 ha)	557	584	584	566	557	315	330	347	328	315
Irrigated land	1194	1194	1194	1194	1194	1194	1194	1194	1194	1194
Dryland	2769	2769	2769	2769	2769	2769	2769	2769	2769	2769
Livestock (LSU)										
Animals										
Water use (Mm <sup>3</sup> )	3874	4031	4031	3549	3874	2475	2566	2564	2280	2475
Agriculture	3552	3709	3709	3227	3552	2176	2244	2242	1958	2176
Urban	322	322	322	322	322	322	322	322	322	322
Streamflow at the river mouth	9272	9160	9160	9290	9272	5406	5341	5342	5416	5406
Nitrogen emissions (1000 tNO <sub>3</sub> -N)										
At the source	236	229	160	234	115	225	220	189	224	105
Entering water	94	91	66	93	46	89	87	73	89	42
bodies										
NO <sub>3</sub> <sup>-</sup> concentration (mg/l NO <sub>3</sub> <sup>-</sup> )										
Ebro River mouth	11.3	11.0	7.7	11.1	5.5	18.4	18.2	15.7	18.3	8.6
GHG emissions (MtCO <sub>2</sub> e)	7.15	6.96	6.85	7.11	6.65	6.93	6.79	6.81	6.92	6.43
Private benefits (M€)										
Agriculture	1925	1970	1937	1937	1642	1727	1764	1772	1761	1444
Urban	1859	1859	1859	1859	1859	1859	1859	1859	1859	1859
Total	3.784	3829	3796	3796	3.501	3586	3623	3623	3620	3303
Env. damages (M€)	409	397	300	406	326	394	386	312	393	312
Social benefits (M€)										
Agriculture	1516	1573	1672	1531	1316	1333	1378	1452	1418	1133
Urban	1859	1859	1859	1859	1859	1859	1859	1859	1859	1859
Total	3375	3432	3531	3390	3175	3192	3237	3311	3277	2292

### 4.2.3. Irrigation modernization

Modernization investments involve upgrading irrigation technologies, which enhance the efficiency of water use and reduce nitrate and GHG emissions. Modernization increases cultivated land to 566,000 ha after substituting surface irrigation by sprinkler and drip systems. However, advanced irrigation systems reduce water withdrawals to 3173 Mm<sup>3</sup> and nitrogen fertilization to 85,000 tN, increasing the efficiency of water and nitrogen use. Therefore, nitrate loads at the source and nitrate concentration at the Ebro mouth are reduced, while the streamflow at the mouth increases. N<sub>2</sub>O emissions also decrease to 0.72 MtCO<sub>2</sub>e. This shows that modernization generates suitable tradeoffs between streamflow, nitrate concentrations and GHG emissions. Advanced irrigation technologies increase yields and farmers' benefits, but modernization costs are very high. As a consequence, the private benefits of irrigation decrease but they are still advantageous compared with the baseline.

Under drought, modernization reduces water use, nitrogen leached, and GHG emissions, increasing social benefits by 85 M $\in$  compared to drought without policies. Although modernization increases streamflow at the mouth, the abatement of nitrate concentration is very small, which shows the tradeoff of this policy between water quantity and quality (Table 3).

# 4.2.4. Manure treatment plants

Manure treatment plants reduce direct and indirect nitrogen loads into watercourses and nitrous oxide emissions into the atmosphere from manure management. These abatement technologies involve high investment, operation and maintenance costs. This study considers plants of 50,000 m<sup>3</sup>/year with nitrification and denitrification processes, with total cost at 7  $\notin$ /m<sup>3</sup> of manure (Flotats et al., 2011). Results under normal flow and drought conditions show that the installation of manure treatment plants maintains water withdrawals by agriculture and streamflow at the river mouth, but achieves significant abatement of both nitrate concentration at the Ebro mouth (by more than half to 5.5 and 8.6 mg/1 NO<sub>3</sub><sup>-</sup>, respectively for normal and drought years) and GHG emissions (down to 6.65 and 6.43 MtCO<sub>2</sub>e, respectively). Environmental damages are curbed by around 80 M€ but the costs of this policy are close to 280 M€, reducing both private and social benefits (Table 3). The investments in manure treatment plants would be reasonable for higher social carbon costs above the current estimates of 40  $\notin$ /tCO<sub>2</sub>e, or for river reaches where highly valuable aquatic ecosystems are damaged by nitrates. Also, manure treatment plants could be the only alternative in areas generating large quantities of manure that cannot be reused as fertilizer because of the lack of cropland in the surroundings.

# 5. Discussion

This research provides a comprehensive analysis of water allocation and agricultural nonpoint pollution in the Ebro basin under normal and drought events, together with the relationship between water quantity and quality. Drought conditions reduce agricultural withdrawals and pollution loads to water media and the atmosphere, although nitrate concentrations increase because of the substantial fall in stream flows. Yang et al. (2015) indicate that these tradeoffs between water quantity and quality are important in considering sustainable development outcomes.

The results on water allocation and agricultural pollution loads during normal weather and droughts provide useful information for decision making. Climate impacts would undermine the sustainability of water systems in the Ebro under current management practices, threatening both irrigated agriculture and environmental flows. The results of drought scenarios call for decisive policy interventions by local, state and federal stakeholders to reduce the vulnerability of the economic sectors, and also to protect the natural environment. This research evaluates several policies relevant for regional and basin water planning. These policies promote the efficient use of water and nutrients, enhance farming conditions and environmental outcomes, and increase farmer's income in some cases. Successful policy implementation and enforcement entail the involvement of water stakeholders in water planning, along with the general public support that would motivate political representatives.

Several policy initiatives have been taken in some countries to address the abatement of agricultural nonpoint pollution, such as the European Nitrates Directive (European Commission, 1991), limiting nitrogen emissions from farming systems to protect groundwater and surface waterways. The purpose is to reduce nitrate pollution into water bodies caused by excessive nitrogen fertilization and manure surplus. However, the achievements of the Nitrates Directive during the last three decades are questionable because the entry of nitrogen in soils has not been curtailed.<sup>4</sup> The main problems with the Directive are that the use of homogeneous measures across very heterogeneous European regions in terms of pollution loads, and the flimsy enforcement mechanism based on penalizing agricultural subsidies (Albiac et al., 2020). Another case is the conservation programs in the United States for reducing agricultural nonpoint pollution. Despite spending 5 billion US dollars per year in conservation programs over the last two decades, there is no clear general improvement of water quality in basins (Ribaudo, 2015).

Our results indicate that the selected policies contribute to the abatement of nonpoint pollution, and improve both water and air quality. The results reveal the tradeoffs and synergies between economic and environmental effects of these abatement policies. Nitrogen optimization (P1), manure fertilization (P2) and irrigation modernization (P3) are interesting policies that reduce polluting emissions into the atmosphere and watercourses, while enhancing the private benefits of farmers. Those policies deliver synergies between the economic and environmental outcomes. However, manure treatment plants (P4) deliver a strong reduction of nonpoint pollution and environmental damages, but they also reduce private benefits because of the high investment and operating costs. This reduction in farmers' income indicates that the uptake of this policy by farmers would be challenging, requiring strong command and control measures coupled with public incentives or subsidies. Drought conditions limit the effectiveness of pollution abatement policies compared with normal weather. However, these policies still have significant economic and environmental positive effects compared to drought conditions without policies. The analysis of mitigation policies supports decision making and contribute to the ongoing policy discussion for designing basin wide sustainable water management.

The use of manure as fertilizer is an effective policy to cut back nitrate concentration, improving water and atmosphere quality (Baccour et al., 2021). According to Strokal et al. (2020), incorporating manure as crop fertilizer is an effective strategy for drastically reduce eutrophication. This policy is considered an important solution to prevent the entry of nitrogen in soils by substituting synthetic fertilizers (Khan and Chang, 2018; Ma et al., 2019; MOA, 2018). Moreover, manure fertilization is quite interesting in the Ebro Basin, especially in Aragon, because the volume of available manure in the region can meet all nitrogen requirements by crops (Orús, 2006). Albiac et al. (2016) indicate that the use of organic fertilizers in Europe could decrease the use of synthetic fertilizers by almost half, thus reducing nitrous oxide emissions and nitrogen loads in watercourses, which would generate around 5200 M€ in environmental benefits. Dalgaard et al. (2014) indicate the successful implementation of this policy in Denmark, with a mix of command and control (fines) and institutional instruments, by showing farmers that substitution of synthetic fertilizers with manure was profitable.

Another interesting policy is irrigation modernization, which enhances water efficiency at parcel level and abates pollution loads. According to Borrego-Marín and Berbel (2019), the impact of irrigation modernization on improved water quality is significant at the basin scale and the implementation of this strategy minimizes nitrogen leaching into water bodies, while providing economic benefits similar to our results. Garcia-Garizábal and Causapé (2010) estimate a 20% reduction in leached nitrogen following the adoption of water conservation measures in an irrigation district in the Ebro. Albiac et al. (2017) indicate that irrigation modernization in Spain could reduce GHG emissions by 2.1 MtCO<sub>2</sub>e, but involves quite high investment costs. Grafton et al. (2018) emphasize the paradox of irrigation efficiency, which indicates that

advanced irrigation technologies increases irrigation efficiency at district level, but could also increase water consumption in the basin. Gains in irrigation efficiency promote more water-intensive crops, double crops or irrigated land expansion, resulting in higher evapotranspiration and lower return flows to watersheds. To avoid the paradox, modernization projects of irrigation districts should include water balances that prevent increases in evapotranspiration.

The choice of policies depends on the objectives of decision makers, but also on the availability of biophysical and economic information. The uptake of policies is related to their cost-efficiency, acceptability by stakeholders, appropriate design of implementation and enforcement mechanisms, and resulting transaction costs. Successful implementation requires effective policies that are socially viable and include appropriate enforcement mechanisms ensuring compliance by stakeholders. A mix of command and control, economic and institutional instruments are needed to facilitate the implementation of sustainable water management. Better education is also important, as seen by the Science Technology Backyards initiative in China, in which scientists, students, and farmers exchange their expertise. In other terms, collective action and cooperation among farmers, policymakers, scientists, and other stakeholders are needed to achieve sustainable policies (Jiao et al., 2016). Overall, implementing cost-effective management strategies requires the successful deployment and uptake of policies and technology packages by stakeholders, as well as organizing their active cooperation.

Our study is novel in two key aspects: First, an optimization model is used to analyze the tradeoffs between water quantity and quality in order to maximize the social benefits of water from agricultural activities and urban centers. Second, the evaluation of nutrient pollution into watercourses and GHG emissions into the atmosphere from irrigated, dryland, and livestock activities under normal and severe droughts conditions. The evaluation of selected policies with the model provides clues on suitable combinations of mitigation and adaptation policies for water and air quality enhancement.

A certain number of simplifying assumptions have been used in developing the hydroeconomic model. The model includes a reduced form hydrological framework, which does not include reservoirs and their linkages with stream flows. Moreover, the model is static and does not include dynamic aspects regarding water allocations, basin stream flows, and drought events. This may change the effectiveness of mitigation and adaptation policies over a multi-year horizon. Despite these limitations, the hydroeconomic model is a good analytical tool to assess the effects of drought scenarios under selected mitigation and adaptation policies for enhancing water allocation and curbing water and air pollution.

Future work could address model improvements such as incorporating significant additional biophysical processes (pollution transport and fate processes, other pollutants), and including water storage of reservoirs and hydropower generation. Other improvements are considering the headwater inflow variables stochastic, modifying the time step of the model from yearly to monthly, and improving the model calibration and validation. The introduction of stochastic variables would be an interesting advance for a better representation of droughts and climate change. This will improve the estimation of nonpoint pollution loads into water streams for a better assessment of policies. Another important aspect that could be included in the analysis is the strategic behavior of stakeholders' in order to figure out the acceptability and stability of cooperative solutions for the abatement of water pollution loads and GHG emissions.

# 6. Conclusions

Water availability and agricultural nonpoint pollution in the Ebro River are analyzed under normal and drought conditions using an integrated hydroeconomic model. The study analyzes a set of mitigation and adaptation policies to address water scarcity and quality, and emissions of greenhouse gases. Results indicate that drought conditions

<sup>&</sup>lt;sup>4</sup> Examples of the limited success of the Nitrates Directive is the Seine River where nitrate pollution at the mouth has doubled since 1991 (Romero et al., 2016), the Po River where nitrate trends have been increasing (Musacchio et al., 2020), and the Thames River where nitrate pollution has not decreased since the 1990s (Howden et al., 2011).

reduce crops with low profitability and high water requirements, raising nitrate concentrations by up to 63% and highlighting the tradeoff between nitrate concentrations and water availability. The assessment of mitigation and adaptation policies provides insights on the synergies and tradeoffs between environmental and economic objectives, as well as on the potential tradeoffs between water quantity and water quality. All evaluated policies improve water quality and reduce the emissions of greenhouse gases. However, the most cost-effective policies are the reduction of nitrogen fertilization, the substitution of synthetic fertilization by manure, and the improvement of irrigation technologies. These cost-effective policies would facilitate the achievement of sustainable water management goals in the basin. Our study could support the decision-making process by contributing to the ongoing policy discussions for the design of basin wide sustainable policies. The findings in the Ebro could have interest also for other rivers basin, especially in arid and semiarid regions with similar agricultural and climate conditions.

# CRediT authorship contribution statement

**Safa Baccour:** Conceptualization, Formal analysis, Investigation, Methodology, Writing – original draft. **Jose Albiac:** Conceptualization, Formal analysis, Investigation, Methodology, Funding acquisition, Supervision, Validation, Writing – review & editing. **Taher Kahil:** Formal analysis, Methodology, Validation. **Encarna Esteban:** Conceptualization, Investigation, Writing – review & editing. **Daniel Crespo:** Conceptualization, Formal analysis, Methodology, Data curation. **Ariel Dinar:** Conceptualization, Methodology, Writing – review & editing.

#### Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

# Acknowledgements

This research has been funded by the INIA RTA 2017-00082-00-00 project of the Spanish Ministry of Science and Innovation, which also provided a doctoral scholarship to Safa Baccour. We also thank Daniel Isidoro (CITA-DGA) for support on biophysical data and modeling of the Ebro Basin.

# Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.jclepro.2021.129459.

# References

- Aein, R., Alizadeh, H., 2021. Integrated hydro-economic modeling for optimal design of development scheme of salinity affected irrigated agriculture in Helleh River Basin. Agric, Water Manag, 243, 106505. https://doi.org/10.1016/j.agwat.2020.106505.
- Ahmadi, B., Moradkhani, H., 2019. Revisiting hydrological drought propagation and recovery considering water quantity and quality. Hydrol. Process. 33 (10), 1492–1505.
- Albiac, J., Calvo, E., Esteban, E., Kahil, T., 2020. The challenge of irrigation water pricing in the Water Framework Directive. Water Altern. (WaA) 13 (3), 674–690.
- Albiac, J., Kahil, T., Notivol, E., Calvo, E., 2017. Agriculture and climate change: potential for mitigation in Spain. Sci. Total Environ. 592, 495–502. https://doi.org/ 10.1016/j.scitotenv.2017.03.110.
- Albiac, J., Kahil, T., Calvo, E., Crespo, D., Esteban, E., 2016. Evaluación Socio-Económica del Proyecto "LIFE-MANEV: Gestión y Tecnologías de Tratamiento de Estiércol para la Protección Medioambiental y la Sostenibilidad de la Ganadería en Europa". Working Document 16-02. CITA-DGA. Zaragoza.
- Arbués, F., Barberán, R., Villanúa, I., 2004. Price impact on urban residential water demand: a dynamic panel data approach. Water Resour. Res. 40, W11402. https:// doi.org/10.1029/2004WR003092.
- Arbués, F., García-Valiñas, M.Á., Villanúa, I., 2010. Urban water demand for service and industrial use: the case of zaragoza. Water Resour. Manag. 24, 4033–4048. https:// doi.org/10.1007/s11269-010-9645-5.

- Babel, M.S., Gupta, A.D., Nayak, D.K., 2005. A model for optimal allocation of water to competing demands. Water Resour. Manag. 19, 693–712. https://doi.org/10.1007/ s11269-005-3282-4.
- Baccour, S., Albiac, J., Kahil, T., 2021. Cost-effective mitigation of greenhouse gas emissions in the agriculture of Aragon, Spain. Int. J. Environ. Res. Publ. Health 18 (3), 1084. https://doi.org/10.3390/ijerph18031084.
- Barbieri, M., Ricolfi, L., Vitale, S., Muteto, P., Nigro, A., Sappa, G., 2019. Assessment of groundwater quality in the buffer zone of limpopo national park, gaza province, southern Mozambique. Environ. Sci. Pollut. Res. 26, 62–77. https://doi.org/ 10.1007/s11356-018-3474-0.
- Berthet, A., Vincent, A., Fleury, P., 2021. Water quality issues and agriculture: an international review of innovative policy schemes. Land Use Pol. 109, 105654. https://doi.org/10.1016/j.landusepol.2021.105654.
- Bekchanov, M., Ringler, C., Bhaduri, A., Jeuland, M., 2015. How would the Rogun Dam affect water and energy scarcity in Central Asia? Water Int. 40, 856–876. https://doi. org/10.1080/02508060.2015.1051788.
- Bekchanov, M., Sood, A., Pinto, A., Jeuland, M., 2017. Systematic review of watereconomy modeling applications. J. Water Resour. Protect. 143, 8. https://doi.org/ 10.1061/(ASCE)WR.1943-5452.0000793.
- Bluemling, B., Wang, F., 2018. An institutional approach to manure recycling: conduit brokerage in Sichuan Province, China. Resour. Conserv. Recycl. 139, 396–406. https://doi.org/10.1016/j.resconrec.2018.08.001.
- Booker, J., Howitt, R., Michelsen, A., Young, R., 2012. Economics and the modeling of water resources and policies. Nat. Res.Model. 25, 168–218. https://doi.org/ 10.1111/i.1939-7445.2011.00105.x.
- Booker, J.F., Michelsen, A.M., Ward, F.A., 2005. Economic impact of alternative policy responses to prolonged and severe drought in the Rio Grande Basin. Water Resour. Res. 41, W02026. https://doi.org/10.1029/2004WR003486.
- Booker, J.F., Young, R.A., 1991. Economic Impacts of Alternative Water Allocation Institutions in the Colorado River Basin. Colo. Colorado Water Resources Research Institute, Colorado State University, Fort Collins.
- Booker, J.F., Young, R.A., 1994. Modeling intrastate and interstate markets for Colorado river water resources. J. Environ. Econ. Manag. 26, 66–87. https://doi.org/10.1006/ jeem.1994.1005.
- Brown, T., Harding, B., Payton, E., 1990. Marginal economic value of streamflow: a case study for the Colorado River Basin. Water Resour. Res. 26, 2845–2859. https://doi. org/10.1029/WR026i012p02845.
- Borrego-Marín, M.M., Berbel, J., 2019. Cost-benefit analysis of irrigation modernization in guadalquivir river basin. Agric. Water Manag. 212, 416–423. https://doi.org/ 10.1016/j.agwat.2018.08.032.
- Breitburg, D., Levin, L.A., Oschlies, A., Grégoire, M., Chavez, F.P., Conley, D.J., Garçon, V., Gilbert, D., Gutiérrez, D., Isensee, K., Jacinto, G.S., Limburg, K.E., Montes, I., Naqvi, S.W.A., Pitcher, G.C., Rabalais, N.N., Roman, M.R., Rose, K.A., Seibel, B.A., Telszewski, M., Yasuhara, M., Zhang, J., 2018. Declining oxygen in the global ocean and coastal waters. Sci 359, eaam7240. https://doi.org/10.1126/ science.aam7240.
- Büntgen, U., Urban, O., Krusic, P., Rybníček, M., Kolář, T., Kyncl, T., Ač, A., Koňasová, E., Čáslavský, J., Esper, J., Wagner, S., Saurer, M., Tegel, W., Dobrovolný, P., Cherubini, P., Reinig, F., Miroslav, T., 2021. Recent European drought extremes beyond Common Era background variability. Nat. Geosci. 1–7 https://doi.org/10.1038/s41561-021-00698-0.
- Buendia, C., Batalla, R.J., Sabater, S., Palau, A., Marcé, R., 2016. Runoff trends driven by climate and afforestation in a Pyrenean Basin. Land Degrad. Dev. 27 (3), 823–838.
- Cai, X., McKinney, D., Lasdon, L., 2003. Integrated hydrologic-agronomic-economic model for river basin management. J. Water Resour. Protect. 129 (1) https://doi. org/10.1061/(ASCE)0733-9496(2003)129:1(4).
- Cai, X., McKinney, D.C., Lasdon, L.S., 2002. A framework for sustainability analysis in water resources management and application to the Syr Darya Basin: sutainability analysis in water resources management. Water Resour. Res. 38, 21. https://doi.org/ 10.1029/2001WR000214, 1-21–14.
- Carolus, J.F., Bartosova, A., Olsen, S.B., Jomaa, S., Veinbergs, A., Zilāns, A., et al., 2020. Nutrient mitigation under the impact of climate and land-use changes: a hydroeconomic approach to participatory catchment management. J. Environ. Manag. 271, 110976. https://doi.org/10.1016/j.jenvman.2020.110976.
- Cammalleri, C., Naumann, G., Mentaschi, L., Formetta, G., Forzieri, G., Gosling, S., Bisselink, B., De Roo, A., Feyen, L., 2020. Global Warming and Drought Impacts in the EU. Publications Office of the European Union, Luxembourg.
- CEDEX, 2020. Anuario de Aforos 2010–2018. MITECO, Madrid. https://ceh.cedex.es/ anuarioaforos/default.asp.
- CHE, 2015. Plan Hidrológico de la Demarcación Hidrográfica del Ebro. Memoria CHE, MAGRAMA, Zaragoza.
- CHE, 2016. Plan Hidrológico de la Demarcación Hidrográfica del Ebro. Memoria CHE, MAGRAMA, Zaragoza.
- CHE, 2020. Informe impress. Análisis de presiones, impactos y riesgos. MITECO, Zaragoza.
- Crespo, D., Albiac, J., Kahil, T., Esteban, E., Baccour, S., 2019. Tradeoffs between water uses and environmental flows: a hydroeconomic analysis in the Ebro basin. Water Resour. Manag. 33, 2301–2317. https://doi.org/10.1007/s11269-019-02254-3.
- Dalgaard, T., Hansen, B., Hasler, B., Hertel, O., Hutchings, N.J., Jacobsen, B.H., Jensen, L.S., Kronvang, B., Olesen, J.E., Schjorring, J.K., 2014. Policies for agricultural nitrogen management—trends, challenges and prospects for improved efficiency in Denmark. Environ. Res. Lett. 9 (11), 115002.
- Dagnino, M., Ward, F., 2012. Economics of agricultural water conservation: empirical analysis and policy implications. Int. J. Water Resour. Dev. 28, 577–600. https://doi. org/10.1080/07900627.2012.665801.

#### S. Baccour et al.

Dasgupta, P., 2021. The Economics of Biodiversity: the Dasgupta Review. HM Treasury, London.

Daudén, A., Teresa, M., Siegler, C., 2011. Proyecto demostrativo de gestión colectiva del purín en Aragón. Sociedad de Desarrollo Medioambiental de. Aragón, Zaragoza.

- Divakar, L., Babel, M.S., Perret, S.R., Gupta, A.D., 2011. Optimal allocation of bulk water supplies to competing use sectors based on economic criterion – an application to the Chao Phraya River Basin, Thailand. J. Hydrol. 401, 22–35. https://doi.org/10.1016/ j.jhydrol.2011.02.003.
- Dogan, M.S., Fefer, M.A., Herman, J.D., Hart, Q.J., Merz, J.R., Medellín-Azuara, J., Lund, J.R., 2018. An open-source Python implementation of California's hydroeconomic optimization model. Environ. Model. Software 108, 8–13. https:// doi.org/10.1016/j.envsoft.2018.07.002.
- Escriva-Bou, A., Lund, J.R., Pulido-Velazquez, M., Hui, R., Medellín-Azuara, J., 2018. Developing a water-energy-GHG emissions modeling framework: insights from an application to California's water system. Environ. Model. Software 109, 54–65. https://doi.org/10.1016/j.envsoft.2018.07.011.
- European Environment Agency, 2018. European Waters: Assessment of Status and Pressures 2018 (EEA Report No 7). European Environment Agency. https://doi.org/ 10.2800/303664.

European Commission, 1991. Council Directive of 12 December 1991 concerning the protection of waters against pollution caused by nitrates from agricultural sources (91/676/EEC) off. J. Eur. Commun. L 375 (1991), 1–8.

- Flotats, X., Foged, H., Blasi, A., Palatsi, J., Magri, A., Schelde, K., 2011. Future Trends on Manure Processing Activities in Europe. Technical Report No. II to the European Commission. Directorate-General Environment, Brussels.
- Garcia-Garizábal, I., Causapé, J., 2010. Influence of irrigation water management on the quantity and quality of irrigation return flows. J. Hydrol. 385 (1–4), 36–43. https:// doi.org/10.1016/j.jhydrol.2010.02.002.
- Grafton, R.Q., Williams, J., Perry, C.J., Molle, F., Ringler, C., Steduto, P., Udall, B., Wheeler, S.A., Wang, Y., Garrick, D., Allen, R.G., 2018. The paradox of irrigation efficiency. Sci 361, 748–750. https://doi.org/10.1126/science.aat9314.

Greve, P., Kahil, T., Mochizuki, J., Schinko, T., Satoh, Y., Burek, P., Fischer, G., Tramberend, S., Burtscher, R., Langan, S., Wada, Y., 2018. Global assessment of water challenges under uncertainty in water scarcity projections. Nat. Sustain. 1, 486–494. https://doi.org/10.1038/s41893-018-0134-9.

- Gunawardena, A., White, B., Hailu, A., Wijeratne, E.M.S., Pandit, R., 2018. Policy choice and riverine water quality in developing countries: an integrated hydro-economic modelling approach. J. Environ. Manag. 227, 44–54. https://doi.org/10.1016/j. jenvman.2018.08.065.
- Harou, J., Pulido-Velazquez, M., Rosenberg, D., Medellin-Azuara, J., Lund, J., Howitt, R., 2009. Hydro-economic models: concepts, design, applications, future prospects. J. Hydrol. 375, 627–643. https://doi.org/10.1016/j.jhydrol.2009.06.037.
- Heinz, I., Pulido-Velazquez, M., Lund, J., Andreu, J., 2007. Hydro-economic modeling in river basin management: implications and applications for the european water framework directive. Water Resour. Manag. 21, 1103–1125. https://doi.org/ 10.1007/s11269-006-9101-8.
- Hernández, N., Gil, M., Garrido, A., Rodríguez, R., 2013. La sequía 2005-2008 en la cuenca del Ebro: Vulnerabilidad, impactos y medidas de gestión. CEIGRAM. Universidad Politécnica de Madrid, Madrid.
- Herrero, A., Gutiérrez-Cánovas, C., Vigiak, O., Lutz, S., Kumar, R., Gampe, D., et al., 2018. Multiple stressor effects on biological quality elements in the Ebro River: present diagnosis and predicted responses. Sci. Total Environ. 630, 1608–1618. https://doi.org/10.1016/j.scitotenv.2018.02.032.
- Howden, N., Burt, T., Worrall, F., Mathias, S., Whelan, M., 2011. Nitrate pollution in intensively farmed regions: what are the prospects for sustaining high-quality groundwater? Water Resour. Res. 47, W00L02. https://doi.org/10.1029/ 2011WR010843.
- IPCC, 2007. In: Metz, B., Davidson, O.R., Bosch, P.R., Dave, R., Meyer, L.A. (Eds.), Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- IPCC, 2019a. 2019 refinement to the 2006 IPCC guidelines for national greenhouse gas inventories, 4. Agriculture, Forestry and Other Land Use. (Chapter 11): N<sub>2</sub>O Emissions from managed soils, and CO<sub>2</sub> emissions from lime and urea application. Geneva, Switzerland.

IPCC, 2019b. 2019 refinement to the 2006 IPCC guidelines for national greenhouse gas inventories, 4. Agriculture, Forestry and Other Land Use. (Chapter 10): Emissions from livestock and manure management. Geneva, Switzerland.

Jiao, X., Lyu, Y., Wu, X., Li, H., Cheng, L., Zhang, C., Yuan, L., Jiang, R., Jiang, B., Rengel, Z., et al., 2016. Grain production versus resource and environmental costs: towards increasing sustainability of nutrient use in China. J. Exp. Bot. 67, 4935–4949. https://doi.org/10.1093/jtb/erw282.

- Jury, W.A., Vaux, H., 2005. The role of science in solving the world's emerging water problems. Proc. Natl. Acad. Sci. Unit. States Am. 102, 15715–15720. https://doi. org/10.1073/pnas.0506467102.
- Kahil, T., Dinar, A., Albiac, J., 2015. Modeling water scarcity and droughts for policy adaptation to climate change in arid and semiarid regions. J. Hydrol. 522, 95–109. https://doi.org/10.1016/j.jhydrol.2014.12.042.

Kahil, T., Ward, F.A., Albiac, J., Eggleston, J., Sanz, D., 2016a. Hydro-economic modeling with aquifer–river interactions to guide sustainable basin management. J. Hydrol 539, 510–524. https://doi.org/10.1016/j.jhydrol.2016.05.057.

Kahil, T., Albiac, J., Dinar, A., Calvo, E., Esteban, E., Avella, L., Garcia-Molla, M., 2016b. Improving the performance of water policies: evidence from drought in Spain. Water 8, 34. https://doi.org/10.3390/w8020034.

Kahil, T., Parkinson, S., Satoh, Y., Greve, P., Burek, P., Veldkamp, T.I., et al., 2018. A continental-scale hydroeconomic model for integrating water-energy-land nexus solutions. Water Resour. Res. 54, 7511–7533. https://doi.org/10.1029/2017WR022478.

Kanter, D.R., Wentz, J.A., Galloway, J.N., Moomaw, W.R., Winiwarter, W., 2017. Managing a forgotten greenhouse gas under existing U.S. law: an interdisciplinary analysis. Environ. Sci. Pol. 67, 44–51. https://doi-org./10.1016/j.envsci.2016.11.00

Khan, M., Chang, Y.-C., 2018. Environmental challenges and current practices in China—a thorough analysis. Sustainability 10 (7), 2547. https://doi.org/10.3390/ su10072547.

- Lassaletta, L., García-Gómez, H., Gimeno, B.S., Rovira, J.V., 2009. Agriculture-induced increase in nitrate concentrations in stream waters of a large Mediterranean catchment over 25 years (1981–2005). Sci. Total Environ. 407, 6034–6043. https:// doi.org/10.1016/j.scitotenv.2009.08.002.
- Lassaletta, L., Romero, E., Billen, G., Garnier, J., García-Gómez, H., Rovira, J.V., 2012. Spatialized N budgets in a large agricultural Mediterranean watershed: high loading and low transfer. Biogeosciences 9, 57–70. https://doi.org/10.5194/bg-9-57-2012.
- Ma, L., Bai, Z., Ma, W., Guo, M., Jiang, R., Liu, J., Oenema, O., Velthof, G.L., Whitmore, A., Crawford, J., 2019. Exploring future food provision scenarios for China. Environ. Sci. Technol. 53 (3), 1385–1393. https://doi.org/10.1021/acs. est.8b04375.
- Martínez, Y., Albiac, J., 2006. Nitrate pollution control under soil heterogeneity. Land Use Pol. 23, 521–532. https://doi.org/10.1016/j.landusepol.2005.05.002.

Myers, S.S., Patz, J.A., 2009. Emerging threats to human health from global environmental change. Annu. Rev. Environ. Resour. 34, 223–252. https://doi.org/ 10.1146/annurev.environ.033108.102650.

- MOA, 2018. Utilization of livestock and poultry manure resources. Action Plan for 2017-2020. Ministry of Agriculture and Rural Affairs of the People's Republic of China. Retrived in March 2019 from. http://m.mof.gov.cn/zcfb/201805/t2018051128928 99.htm.
- Moraes, M., Cai, X., Ringler, C., Albuquerque, B., da Rocha, S., Amorim, C., 2010. Joint water quantity-quality management in a biofuel production area—integrated economic-hydrologic modeling analysis. J. Water Resour. Plann. Manag. 136 (4), 502–511. https://doi.org/10.1061/(ASCE)WR.1943-5452.0000049.
- Musacchio, A., Re, V., Mas-Pla, J., Sacchi, E., 2020. EU Nitrates Directive, from theory to practice: environmental effectiveness and influence of regional governance on its performance. Ambio 49, 504–516.
- NOAA, 2021. U.S. Billion-Dollar Weather and Climate Disasters. National Centers for Environmental Information. NOAA. Asheville.

Ollero, A., 2007. Channel adjustments, floodplain changes and riparian ecosystems of the middle Ebro River: assessment and management. Water Resour. Dev. 23 (1), 73–90.

- Orús, F., 2006. Fertilización nitrogenada: guía de actualización. Informaciones Técnicas, Número extraordinario. Centro de Transferencia Agroalimentaria. Dirección General de Desarrollo Rural (Zaragoza).
- Parris, k., 2011. Impact of agriculture on water pollution in OECD countries: recent trends and future prospects. Int. J. Water Resour. Dev. 27 (1), 33–52. https://doi. org/10.1080/07900627.2010.531898.
- Peña-Haro, S., Pulido-Velazquez, M., Sahuquillo, A., 2009. A hydro-economic modelling framework for optimal management of groundwater nitrate pollution from agriculture. J. Hydrol. 373, 193–203. https://doi.org/10.1016/j. ibvdrol.2009.04.024.
- Peña-Haro, S., Pulido-Velazquez, M., Llopis-Albert, C., 2011. Stochastic hydro-economic modeling for optimal management of agricultural groundwater nitrate pollution under hydraulic conductivity uncertainty. Environ. Model. Software 26, 999–1008. https://doi.org/10.1016/j.envsoft.2011.02.010.

Poff, N.L., Allan, J.D., Bain, M.B., Karr, J.R., Prestegaard, K.L., Ritcher, B.D., Sparks, R.E., Stromberg, J.C., 1997. The natural flow regime: a paradigm for river conservation and restoration. Bioscience 47, 769–784. https://doi.org/10.2307/1313099.

Ribaudo, M., 2015. The limits of voluntary conservation programs. Choices 30 (2), 1–5. Ringler, C., Huy, N.V., Msangi, S., 2006. Water allocation policy modeling for the Dong

Nai River basin: an integrated perspective. J. Am. Resour. Assoc. 42, 1465–1482. Ringler, C., Cai, X., 2006. Valuing fisheries and wetlands using integrated economichydrologic modeling—mekong river basin. J. Water Resour. Plann. Manag. 132, 480–487. https://doi.org/10.1061/(ASCE)0733-9496(2006)132:6(480).

Ringler, C., von Braun, J., Rosegrant, M., 2004. Water policy analysis for the mekong river basin. Water Int. 29, 30–42. https://doi.org/10.1080/02508060408691746.

- Rockström, J., Falkenmark, M., Karlberg, L., Hoff, H., Rost, S., Gerten, D., 2009. Future water availability for global food production: the potential of green water for increasing resilience to global change. Water Resour. Res. 45, W00A12. https://doi. org/10.1029/2007wr006767.
- Romero, E., Le Gendre, R., Garnier, J., Billen, G., Fisson, C., Silvestre, M., Riouf, P., 2016. Long-term water quality in the lower Seine: lessons learned over 4 decades of monitoring. Environ. Sci. Pol. 58, 141–154. https://hal.sorbonne-universite.fr/hal -01288154.

Simelton, E., Fraser, E.D.G., Termansen, M.T., Benton, G., Gosling, S.N., et al., 2012. The socioeconomics of food crop production and climate change vulnerability: a global scale quantitative analysis of how grain crops are sensitive to drought. Food Secur 4, 163–179. https://doi.org/10.1007/s12571-012-0173-4.

Smith, Braathen, 2015. Monetary Carbon Values in Policy Appraisal: an Overview of Current Practices and Key Issues. OECD, Environment Working Paper. París.

- Strokal, M., Kahil, T., Wada, Y., Albiac, J., Bai, Z., Ermolieva, T., et al., 2020. Costeffective management of coastal eutrophication: a case study for the yangtze river basin. Resour. Conserv. Recycl. 154, 104635. https://doi.org/10.1016/j. resconrec.2019.104635.
- UNESCO, 2021. Water scarcity and quality. https://en.unesco.org/themes/water-securi ty/hydrology/water-scarcity-and-quality.

#### S. Baccour et al.

Van Vliet, M.T., Flörke, M., Wada, Y., 2017. Quality matters for water scarcity. Nat. Geosci. 10, 800–802.

- Vericat, D., Batalla, R.J., 2006. Sediment transport in a large impounded river: the lower Ebro, NE Iberian Peninsula. Geomorphology 79 (1–2), 72–92. https://doi.org/ 10.1016/j.geomorph.2005.09.017.
- Ward, F.A., 2021. Hydroeconomic analysis to guide climate adaptation plans. Front. Water. 3, 80. https://doi.org/10.3389/frwa.2021.681475.
- Ward, F.A., Pulido-Velazquez, M., 2012. Economic costs of sustaining water supplies: findings from the rio grande. Water Resour. Manag. 26, 2883–2909. https://doi.org/ 10.1007/s11269-012-0055-8.
- Ward, F.A., Pulido-Velazquez, M., 2008. Efficiency, equity, and sustainability in a water quantity-quality optimization model in the Rio Grande basin. Ecol. Econ. 66, 23–37. https://doi.org/10.1016/j.ecolecon.2007.08.018.
- Yang, Y., Yin, L., Zhang, Q., 2015. Quantity versus quality in China's South-to-North Water Diversion Project: a system dynamics analysis. Water 7 (5), 2142–2160. https://doi.org/10.3390/w7052142.